NP02 Operations

- Some Coldbox Bottom Drift Electronics data was taken on Thurs, Fri., and this morning.
- Several fixes to FTS and FTS-Light were needed to get pipeline working again
- Top electronics group plans to start working again this week

NP04 Operations

No report



Production

- Attempted to run high-energy request, failures believed due to non-standard FCL files
- Saw job failures at BNL and JINR, jobs ran OK but failed to copy back with "93 protocol not supported"
- Probe job using earlier version of ifdhc was ok at same sites.
- May be missing gfal plugins, still investigating

Data Management Ops

- Tests with Fermi-based FTS3 server successful and stable
- Fixes of xrootd on np04-srv-xxx machines at EHN-1 to run nonprivileged
- Fixes of FTS config @ CERN to update the hosts we send to.
- Investigating some bad copies over the weekend
- Copied back most of production run at NERSC to date.
- Data Challenge planning to have a planning session in Monday morning May 9, next week.